Author's response to reviews

Title: Acculturation and Gestational Weight Gain in a Predominantly Puerto Rican Population

Authors:

Alison Tovar (alison_tovar@mail.uri.edu)
Lisa Chasan-Taber (lct@schoolph.umass.edu)
Odilia I Bermudez (odilia.bermudez@tufts.edu)
Raymond R Hyatt (raymond.hyatt@tufts.edu)
Aviva Must (Aviva.must@tufts.edu)

Version: 4 Date: 9 November 2012

Author's response to reviews:

We have made the following changes requested:
1. In the abstract corrected 1.0kg to 0.9kg.
2. Included a competing interest section.